Author’s response to reviews

Title: Unplanned readmission rates, length of hospital stay, mortality, and medical costs of ten common medical conditions: a retrospective analysis of Hong Kong hospital data

Authors:

Eliza LY Wong (lywong@cuhk.edu.hk)
Annie WL Cheung (anniewlcheung@cuhk.edu.hk)
Michael CM Leung (cmleung123@hotmail.com)
Carrie HK Yam (carrieyam@cuhk.edu.hk)
Frank WK Chan (cwkfrank@cuhk.edu.hk)
Fiona YY Wong (fionawong@cuhk.edu.hk)
Eng Kiong Yeoh (yeoh_ek@cuhk.edu.hk)

Version: 3 Date: 3 June 2011

Author’s response to reviews: see over
3 June 2011

Editor-in-Chief

BMC Health Service & Research

Dear Sir/Madam,

Re: Manuscript MS: 1987061455482610

“Variations of 10 common medical conditions on unplanned readmission, length of hospital stay, mortality, and medical cost: A retrospective cohort study of hospital episode statistics in Hong Kong”

We are very grateful for your email on 19th May 2011 concerning this manuscript, and the offer to allow us to revise it in accordance with the comments of the editorial team. The paper has been copyedited in response to editor’s suggestion using Edanz services. As the suggestion by the Edanz services, the title of paper is revised to “Unplanned readmission rates, length of hospital stay, mortality, and medical costs of ten common medical conditions: a retrospective analysis of Hong Kong hospital data”

We appreciate this opportunity to resubmit a copyedited manuscript and trust that you and your reviewer will find it sufficiently improved to justify publication in the BMC Health Service & Research. Thank you very much for your kind consideration.

Yours sincerely,
Eliza Wong
Assistant Professor
School of Public Health & Primary Care
The Chinese University of Hong Kong